Author's response to reviews

Title: Prevalence and Factors Associated with Urinary Incontinence in Community Dwelling Adult Women: Implications for Health Promotion

Authors:

Vidya Seshan (vidya69@squ.edu.om)
joshua K Muliira (jmuliira@squ.edu.om)

Version: 2 Date: 20 October 2012

Author's response to reviews: see over
Editor

Re: Re Submission of a Manuscript for Publication

After Compliments

Please kindly receive the re-submission of manuscript titled “Prevalence and Factors Associated with Urinary Incontinence in Community Dwelling Adult Women: Implications for Health Promotion” for consideration as a candidate for publication in the Journal BMC Women’s Health.

As per your request we have made changes in the abstract and in the manuscript we have added the ethical committee of Vinayaka Mission University.

The title page we are included in front of the Manuscript.

We look forward to your feedback and to being published in the reputable BMC Women’s Health.

Kind regards,

Dr. Vidya Seshan,

On behalf of the authors